Author’s response to reviews

Title: Music-instruction Intervention for Treatment of Post-traumatic Stress Disorder: A Randomized Pilot Study

Authors:

Liliana Pezzin (lpezzin@mcw.edu)

Eric Larson (Eric.Larson@va.gov)

William Lorber (William.Lorber@va.gov)

Emily McGinley (mcginley@mcw.edu)

Timothy Dillingham (timothy.dillingham@uphs.upenn.edu)

Version: 4 Date: 27 Nov 2018

Author’s response to reviews:

November 27, 2018

Matthew Hickey

Editor, BMC Psychology

Dear Dr. Hickey,

We are delighted with the conditional acceptance of our manuscript "Music-instruction Intervention for Treatment of Post-traumatic Stress Disorder: A Pilot Study." We have revised the manuscript to ensure its consistency with the guidelines identified in your email dated of 11/14/18, as follows:

1. Abstract headings were changed, as requested.

2. Trial registration number is now available as the last line of Abstract, as requested.

3. We clarified in the Ethics section that participants provided written informed consent.
4. We amended the Funding portion of the Declarations page, as requested.

5. We removed the titles from the figure files and included a Figure Legend section after References, as requested.

6. We eliminated any marks or highlights from the final manuscript.

Once again, thank you for your time and consideration of our manuscript.